Author's response to reviews

Title: IgG-index predicts neurological morbidity in patients with infectious central nervous system diseases

Authors:

Peter Lackner (peter.lackner@i-med.ac.at)
Elif Guengoer (elif.akartuna@student.i-med.ac.at)
Ronny Beer (ronny.beer@i-med.ac.at)
Gregor Broessner (gregor.broessner@i-med.ac.at)
Raimund Helbok (raimund.helbok@uki.at)
Florian Deisenhammer (florian.deisenhammer@i-med.ac.at)
Erich Schmutzhard (erich.schmutzhard@i-med.ac.at)
Bettina Pfausler (b.pfausler@i-med.ac.at)

Version: 4 Date: 8 July 2010

Author's response to reviews:

Dear Professor Norton,

all required formatting changes are completed.
We would like to thank you very much for accepting our paper for publication in BMC Infectious diseases.

Yours sincerely,

Peter Lackner, MD
Bettina Pfausler, MD